Author's response to reviews

Title: Uniform distribution of three Candida albicans microsatellite markers in two French ICU populations supports a lack of nosocomial cross-contamination

Authors:

Odile O Eloy (oeloy@ch-versailles.fr)
Stephanie S Marque (x.juillet@wanadoo.fr)
Francoise F Botterel (botterel@univ-paris12.fr)
Francois F Stephan (francois.stephan@chu-guadeloupe.fr)
Jean-Marc J-M Costa (jean-marc.costa@ahparis.org)
Virginie V Lasserre (virginie.lasserre@univ-paris5.fr)
Stephane S Bretagne (bretagne@univ-paris12.fr)

Version: 4 Date: 10 October 2006

Author's response to reviews:

Sentence added page 4 in the methods section:
This observational study did not need approval from the Comite Consultatif de Protection des Personnes dans la Recherche Biomédicale, because it did not modify current diagnostic or therapeutic strategies. Written or oral informed consent was obtained from patients or from their relatives.